Author's response to reviews

Title: X-linked Hyper IgM (HIGM1) in an African kindred: The first report from South Africa

Authors:

Ms Sandra Pienaar (spienaar@ich.uct.ac.za)
Dr Brian S Eley (beley@ich.uct.ac.za)
Jane Hughes (jhughes@ich.uct.ac.za)
Prof Howard E Henderson (henders@ich.uct.ac.za)

Version: 3 Date: 21 Nov 2003

21 November 2003

Dr E Parkin
Editorial Administrator
BMC Journals

Dear Dr Parkin

MS: 6386098622198204
X-linked Hyper IgM (HIGM1) in an African kindred: The first report from South Africa

Herewith, the corrected manuscript. I hope that it meets the Journal's requirements.

Yours sincerely

Dr Brian Eley